Author's response to reviews

Title: The development of a mini-array for estimating the disease states of gastric adenocarcinoma by array CGH

Authors:

Tomoko Furuya (tfuruya@yamaguchi-u.ac.jp)
Tetsuji Uchiyama (tetuuchi@silver.ocn.ne.jp)
Atsushi Adachi (tetuuchi@silver.ocn.ne.jp)
Takae Okada (takae@yamaguchi-u.ac.jp)
Motonao Nakao (motonao@remus.dti.ne.jp)
Atsunori Oga (oga@yamaguchi-u.ac.jp)
Song-Ju Yang (sjyang@yahoo.co.jp)
Shigeto Kawauchi (shig@yamaguchi-u.ac.jp)
Kohsuke Sasaki (kohsuke@yamaguchi-u.ac.jp)

Version: 5 Date: 20 November 2008

Author’s response to reviews: see over
November 20, 2008

Dr. Robin Cassady-Cain  
Assistant Editor  
BMC-series journals

Dear Dr. Cassady-Cain,

Thank you for your e-mail concerning the formatting changes for publication of our paper (MS#1915766214173097). The formatting changes and proofreading has been completed according to your indications. We are looking forward to reading our paper on BMC Cancer.

Thank you for your cooperation.

Best regards,

Kohsuke Sasaki, M.D.
Responses to comments and suggestions from the editor

Major changes

Author list- We’ve changed the list of authors in the manuscript as they are in the submission system.

Reference 17-We’ve already inserted this reference in “Array CGH analysis” in “Methods” of our manuscript.

Additional files-We added a section titled “Additional files” at the end of the manuscript.

Minor changes

We removed ‘Furuya, et al. ‘ form the top of each page.

We removed short title and key words.

We listed the e-mail addresses for all authors on the title page.